Reviewer #1:
**Scientific Quality**: Grade D (Fair)
**Language Quality**: Grade B (Minor language polishing)
**Conclusion**: Minor revision
**Specific Comments to Authors**: 1- There are typos and grammatical errors that need to be corrected. 2- It is recommended that authors put figures together in multiple figure panels.

Response: Special thanks to you for your good comments. We have corrected typos and grammatical errors, and have put figures together in multiple figure panels according to the Reviewer’s suggestion.

Reviewer #2:
**Scientific Quality**: Grade C (Good)
**Language Quality**: Grade B (Minor language polishing)
**Conclusion**: Minor revision
**Specific Comments to Authors**: This case report shows some evidence of Vedolizumab in the treatment of immune checkpoint inhibitor-induced colitis. It will be helpful in clinic. But there are some issues, the author should address to improve their manuscript. 1. Recently a study has been reported Vedolizumab treatment for immune checkpoint inhibitor-induced enterocolitis. PMID: 28204866, please talk about the association in the discussion. 2. The label of Figure H, that’s H not F in the picture. 3. Please confirm the magnification of picture in pathological images is the eyepiece multiply by the objective, or add the scale bar. 4. The resolution and color background of some colonoscopy photos are not good, please try to improve them.

Response: Special thanks to you for your good comments. We have re-written this article according to the Reviewer’s suggestion.